Author's response to reviews

Title: Role of 14-3-3-sigma in poor prognosis and in radiation and drug resistance of human pancreatic cancers

Authors:

Zhaomin Li (li5@iupui.edu)
Zizheng Dong (zidong@iupui.edu)
David Myer (dlmyer@iupui.edu)
Michele Yip-Schneider (myipschn@iupui.edu)
Jianguo Liu (liu54@iupui.edu)
Ping Cui (PCui@dow.com)
C Max Schmidt (maxschmi@iupui.edu)
Jian-Ting Zhang (jianzhan@iupui.edu)

Version: 4 Date: 29 October 2010

Author's response to reviews:

We made the following changes per editorial suggestions:

Affiliations for each author have been updated and linked as suggested.

The table has been recreated using ms word Tables tool.

We have made various corrections of typos and grammar throughout the manuscript.

Figures in jpg have been replaced with that in TIFF format